Author's response to reviews

Title: Treatment of depressive disorders in primary care - protocol of a multiple treatment systematic review of randomized controlled trials

Authors:

Klaus Linde (Klaus.Linde@lrz.tum.de)
Isabelle Schumann (isabelle.schumann@lrz.tum.de)
Karin Meissner (karin.meissner@med.uni-muenchen.de)
Susanne Jamil (susanneohnsmann@gmx.de)
Levente Kriston (l.kriston@uke.uni-hamburg.de)
Gerta Rücker (ruecker@imbi.uni-freiburg.de)
Gerd Antes (antes@cochrane.de)
Antonius Schneider (antonius.schneider@lrz.tum.de)

Version: 3 Date: 8 November 2011

Author's response to reviews: see over
Dear production team,

enclosed we submit a revised version of our manuscript accepted for publication in BMC Family Practice. Below is our point by point response regarding the changes requested.

Major revisions (we require the author to make these changes)
Additional files: Please remove the additional file
Response: Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
Response: Done

Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.
Response: Done

Sincerely
Klaus Linde